Author’s response to reviews

Title: Healthcare professionals' and patients' perspectives on consent to clinical genetic testing: moving towards a more relational approach

Authors:
Gabrielle Samuel (gabbysamuel@gmail.com; G.Samuel@bsms.ac.uk)
Sandi Dheensa (S.Dheensa@soton.ac.uk)
Bobbie Farsides (b.farsides@bsms.ac.uk)
Angela Fenwick (a.j.fenwick@soton.ac.uk)
Anneke Lucassen (a.m.lucassen@soton.ac.uk)

Version: 3 Date: 13 Jul 2017

Author’s response to reviews:

Dear Clare,

Here is a clean copy of the manuscript with the suggested Editorial amendments, including:

1. Replacing all [de-identified] terms in the manuscript with author names
2. Adding a conclusion
3. Moving information about ethics approval to the end of the document.
4. Our supplementary material should be titled 'HCP topic guide and patient interview schedule'

Kind Regards

Gabby